Author's response to reviews

Title: Differential Expression of Cancer Associated Proteins in Breast Milk Based on Age at First Full Term Pregnancy and Length of Nursing

Authors:

Wenyi Qin (wenyi.qin@med.und.edu)
Ke Zhang (ke.zhang@med.und.edu)
beth kliethermes (beth.kliethermes@med.und.edu)
rachel l ruhlen (rachel@ruhlendavis.org)
eva p browne (ebrowne@vasci.umass.edu)
kathleen f arcaro (karcaro@vasci.umass.edu)
edward r sauter (esauter@medicine.nodak.edu)

Version: 2 Date: 10 January 2012

Author's response to reviews: see over
January 10, 2012

Editorial Board
BMC Cancer

RE: Differential Expression of Cancer Associated Proteins in Breast Milk Based on Age at First Full Term Pregnancy

Dear Board Members:

Please find our revised submission of the above referenced manuscript. Below we address each Reviewer's comments point by point. Additions to the text are indicated in red. Deletions are indicated by a strikethrough.

Sincerely,

Edward Sauter

Suggested reviewers:

Dr. Ferdinando Mannello
Ist. di Istologia ed Analisi di Laboratorio
Facoltà di Scienze MM. FF. NN.
Via O. Ubaldini, 7
Università degli Studi "Carlo Bo"
61029 Urbino (PU) - Italia

Tel. 0722-351479
Fax 0722-322370
E-mail  ferdinando.mannello@uniurb.it

Dr. Mary Daly
Fox Chase Cancer Center
7701 Burholme Ave.
Philadelphia, PA 19111
Phone: 215.728.2791
Email: MB_Daly@fccc.edu